Author's response to reviews

Title: Opening the black box of quality improvement collaboratives: an Actor-Network theory approach

Authors:

Tineke Broer (broer@bmg.eur.nl)
Anna P Nieboer (nieboer@bmg.eur.nl)
Roland A Bal (r.bal@bmg.eur.nl)

Version: 3 Date: 31 August 2010

Author's response to reviews: see over
Dear Nina Titmus,

Thank you for accepting our manuscript “Opening the black box of quality improvement collaboratives: an Actor-Network theory approach” for publication in BMC Health Services Research. Based on the reviewer’s suggestion we changed the title of the manuscript, which now contains the term ‘Actor-Network Theory’. Furthermore, we checked the format of the manuscript and made some changes, mainly on the title page.

Sincerely,

Tineke Broer, Anna Nieboer and Roland Bal